Author's response to reviews

Title: Changes in health risk behaviors of elementary school students in northern Taiwan from 2001 to 2003: results from the child and adolescent behaviors in long-term evolution study

Authors:

Wen-Chi Wu (wenchiwu@nhri.org.tw)
Hsing-Yi Chang (hsingyi@nhri.org.tw)
Lee-Lan Yen (leelan@ntu.edu.tw)
Tony Szu-Hsien Lee (tonylee@ntnu.edu.tw)

Version: 4 Date: 30 October 2007

Author's response to reviews:

The Reviewers, BMC Public Health October 30, 2007

Dear Reviewers:

We have made some revisions on request. Please refer to the attachment. Thank you for accepting this paper.

Yours sincerely,

Lee-Lan Yen
Sc. D
Professor
Institute of Health Policy and Management,
College of Public Health,
National Taiwan University.
Room 623, No.17, Hsu-Chow Road, Taipei, Taiwan, R.O.C.
Email: lan@ha.mc.ntu.edu.tw